Author's response to reviews

Title: Protecting the underscreened women in developed countries. The value of HPV test

Authors:

Raquel Ibáñez (raquelip@iconcologia.net)
Josefina Autonell (jautonell@chv.cat)
Montserrat Sardà (msarda@chv.cat)
Nayade Crespo (ncrespo.cc.ics@gencat.cat)
Pilar Piqué (ppique.cc.ics@gencat.cat)
Amparo Pascual (a-pascual@hotmail.com)
Clara Martí (cmartill@fhag.es)
Montserrat Fibla (montsefiblasimo@yahoo.es)
Cristina Gutierrez (cgutierrez.hj23.ics@gencat.cat)
Belén Lloveras (BLloveras@parcdesalutmar.cat)
Judit Moreno-Crespi (crespi81@gmail.com)
Anna Torrent (atorrentesp@gmail.com)
Núria Baixeras (nbaixeras@bellvitgehospital.cat)
María Alejo (maria.alejo@sanitatintegral.org)
F. Xavier Bosch (s.sanjose@iconcologia.net)
Silvia de Sanjose (x.bosch@iconcologia.net)

Version: 3 Date: 31 July 2014

Author's response to reviews: see over
L'Hospitalet de Llobregat (Barcelona), 31 July 2014

MS: 5413359631309702 PROTECTING THE UNDERSCREENED WOMEN IN DEVELOPED COUNTRIES. THE VALUE OF HPV TEST

Dear Editor,

Thank you for your correspondence. Please see below the description of each of the co-authors.

We have uploaded the revised manuscript following the journal instructions where we have added “authors' contributions”. You can read the following paragraph:

RI: was responsible of the coordination for the collection and the quality of clinical data among the different pathology laboratories, had full access to all data, performed the data analysis and drafted all the versions of the manuscript. SdS: was the project leader and designed the study, had full access to all data, performed the data analysis and drafted all the versions of the manuscript. FXB: participated in the study design, contributed to draft the manuscript and made a critical review and proofread of the document. JA, MS, NC, PP, AP, CM, MF, CG, BL, JM, AT, NB, MA: were responsible for the collection and the quality of clinical data among the different pathology laboratories and provided comment to the different drafts. All the authors: read and approved the final version of the manuscript.

I hope that with these changes the manuscript will be finally accepted.

Yours sincerely,

Silvia de Sanjosé MD, PhD
Chief, Unit of Infections & Cancer
Catalan Institute of Oncology